Author's response to reviews

Title: Genomewide Association Study for Onset Age in Parkinson Disease

Authors:

Jeanne C Latourelle (jlatoure@bu.edu)
Nathan Pankratz (npankrat@iupui.edu)
Alexandra Dumitriu (adumitri@bu.edu)
Jemma B Wilk (jwilk@bu.edu)
Stefano Goldwurm (goldwurm@parkinson.it)
Gianni Pezzoli (Pezzoli@parkinson.it)
Claudio B Mariani (mariani@parkinson.it)
Anita L Destefano (adestef@bu.edu)
Cheryl Halter (chalter@iupui.edu)
James F Gusella (gusella@helix.mgh.harvard.edu)
William C Nichols (Bill.Nichols@cchmc.org)
Richard H Myers (rmyers@bu.edu)
Tatiana Foroud (tforoud@iupui.edu)
PSG & PROGENI Investigators and Coordinators [PIC] (tforoud@iupui.edu)
PSG & PROGENI Molecular Genetic Laboratory [PMGL] (tforoud@iupui.edu)
GenePD Investigators and Coordinators [GIC] (rmyers@bu.edu)
GenePD Molecular Genetics Laboratories [GMGL] (rmyers@bu.edu)

Version: 4 Date: 14 September 2009

Author's response to reviews: see over
September 14, 2009

Dear Editors,

We have completed the major and minor formatting changes requested for our manuscript ‘Genomewide association study for onset age in Parkinson disease’.

Please let us know if any additional revisions are required.

Sincerely,

Tatiana Foroud, Ph.D.
(317) 278-1291

Richard Myers, Ph.D.
(617) 638-5376